Author’s response to reviews

Title: Jail-to-Community Treatment Continuum for Adults with Co-Occurring Substance Use and Mental Disorders: Study Protocol for a Pilot Randomized Control Trial

Authors:

Richard Van Dorn (rvandorn@rti.org)
Sarah Desmarais (sdesmarais@ncsu.edu)
Candalyn Rade (cbrade@ncsu.edu)
Elizabeth Burris (enburris@ncsu.edu)
Gary S. Cuddeback (cuddeback@mail.schs.r.unc.edu)
Kiersten Johnson (kijohnson@rti.org)
Stephen Tueller (stueller@rti.org)
Megan Comfort (mcomfort@rti.org)
Kim Mueser (mueser@bu.edu)

Version: 2 Date: 03 Jul 2017

Author’s response to reviews:

1) Please stress throughout the manuscript that this is a pilot RCT (also make this clear in the abstract, title etc..)

   Revised throughout the manuscript.

2) Please state what the primary outcome is or explain why no primary outcome is chosen.

   Done. See page 14.
3) Please mention that moderators of treatment effect are exploratory (there is no sufficient power to identify moderators)

Done. See page 17.

4) Who will allocate participants after randomization (is allocation concealed?)

See pages 12 and 13.